

Make Your Publications Visible.

A Service of



Leibniz-Informationszentrum Wirtschaft Leibniz Information Centre

Li, Xiao; Wang, Yanchao; You, Hong

Article

Executive compensation and conflict between shareholders and creditors: Evidence from creditor litigation

China Journal of Accounting Research

Provided in Cooperation with:

Sun Yat-sen University

Suggested Citation: Li, Xiao; Wang, Yanchao; You, Hong (2020): Executive compensation and conflict between shareholders and creditors: Evidence from creditor litigation, China Journal of Accounting Research, ISSN 1755-3091, Elsevier, Amsterdam, Vol. 13, Iss. 4, pp. 405-424, https://doi.org/10.1016/j.cjar.2020.07.001

This Version is available at: https://hdl.handle.net/10419/241827

Standard-Nutzungsbedingungen:

Die Dokumente auf EconStor dürfen zu eigenen wissenschaftlichen Zwecken und zum Privatgebrauch gespeichert und kopiert werden.

Sie dürfen die Dokumente nicht für öffentliche oder kommerzielle Zwecke vervielfältigen, öffentlich ausstellen, öffentlich zugänglich machen, vertreiben oder anderweitig nutzen.

Sofern die Verfasser die Dokumente unter Open-Content-Lizenzen (insbesondere CC-Lizenzen) zur Verfügung gestellt haben sollten, gelten abweichend von diesen Nutzungsbedingungen die in der dort genannten Lizenz gewährten Nutzungsrechte.

Terms of use:

Documents in EconStor may be saved and copied for your personal and scholarly purposes.

You are not to copy documents for public or commercial purposes, to exhibit the documents publicly, to make them publicly available on the internet, or to distribute or otherwise use the documents in public.

If the documents have been made available under an Open Content Licence (especially Creative Commons Licences), you may exercise further usage rights as specified in the indicated licence.



ND https://creativecommons.org/licenses/by-nc-nd/4.0/





Contents lists available at ScienceDirect

China Journal of Accounting Research

journal homepage: www.elsevier.com/locate/cjar



Executive compensation and conflict between shareholders and creditors: Evidence from creditor litigation



Xiao Li^a, Yanchao Wang^a, Hong You^{b,*}

ARTICLE INFO

Article history: Received 21 October 2019 Accepted 8 July 2020 Available online 29 July 2020

Keywords:
Corporate governance
Agency problem
Creditor litigation
Pay-performance sensitivity

ABSTRACT

Using creditor litigation data from China, we investigate whether creditors can participate in corporate governance when agency conflict between shareholders and creditors is severe. By comparing firms that have experienced creditor lawsuits (litigation firms) with those that have not (non-litigation firms), we find that litigation firms have lower pay-performance sensitivity before lawsuits, suggesting that these firms have weaker corporate governance. This result is consistent with our expectation that creditors participate in corporate governance by introducing external monitoring when internal monitoring, dominated by shareholders, is insufficient. We also find that the association is stronger for firms with more severe shareholder-creditor agency conflict. Moreover, creditor litigation is strongly related to low pay-performance sensitivity when the external legal environment is strong. Our results remain robust to different model specifications and after addressing endogeneity problems. © 2020 Sun Yat-sen University. Production and hosting by Elsevier B.V. This is an open access article under the CC BY-NC-ND license (http://creativecommons.org/licenses/by-nc-nd/4.0/).

1. Introduction

Shareholders and creditors are key financing providers. They invest in corporations to obtain future returns on investment. As they wish to obtain the expected rate of return, shareholders and creditors are concerned about agency problems and corporate governance in the firms in which they invest (Shleifer and Vishny, 1997; Williamson, 1988; Jensen and Murphy, 1990; La Porta et al., 2000; Kroszner and Strahan, 2001; Martynova and Renneboog, 2008; Nini et al., 2012). Regulators have called for the protection of the rights of shareholders

E-mail addresses: afxiaoli@cufe.edu.cn (X. Li), aeroant@126.com (Y. Wang), youhong@swu.edu.cn (H. You).

^a School of Accountancy, Central University of Finance and Economics, China

^b College of Economics and Management, Southwest University, China

^{*} Corresponding author.

and creditors to overcome financing frictions and help firms access external capital (Armour et al., 2015). Therefore, previous studies in economics and finance focus on how to protect the rights of shareholders and creditors through country-level and firm-level monitoring, i.e., legal protection and corporate governance. These two monitoring mechanisms interact; they are either substitutive or complementary (Kim et al., 2007; Gungoraydinoglu and Öztekin, 2011; Abdi and Aulakh, 2012; Ernstberger and Grüning, 2013; El Ghoul et al., 2018; El Ghoul et al., 2017). A lack of country-level legal monitoring increases the likelihood of weak corporate governance, while weak corporate governance requires the intervention of a strong legal environment. Therefore, interaction between the legal system and corporate governance is essential to protect the rights of stakeholders.

However, creditors and shareholders play different roles in corporate governance. It is difficult for creditors to participate in corporate governance, which is dominated by shareholders and management. The literature identifies two types of agency problem in corporations. The first type (Type I) is the standard agent-principal problem between management and shareholders. The second type (Type II) is related to conflicts between shareholders and creditors. In the first type, shareholders use a compensation mechanism to align the interests of management with their own interests. In the second type, creditors participate less directly than shareholders in corporate governance (Nini et al., 2012), because shareholders have residual control rights. The managerial decision-making process is mainly influenced by shareholders, through compensation, shareholder meetings, the voting process, the election of directors, etc. Except in the event of violation of debt covenants, or if the firm defaults or declares bankruptcy, creditors are less directly involved in corporate governance (Townsend, 1979; Gale and Hellwig, 1985; Hart and Moore, 1998; Armour et al., 2015). However, shareholder participation in corporate governance may be insufficient, especially when agency problems are serious. Despite the high agency cost for debtholders, previous studies pay limited attention to how creditors affect corporate governance when shareholders cannot properly reduce moral hazard in the company. Studies suggest that litigation in the form of lawsuits is a critical stopgap measure in corporate governance and stakeholder protection, and is the most frequently used external legal intervention in response to internal management misconduct (Appel, 2019). Therefore, creditor litigation represents the participation of creditors in corporate governance through lawsuits. In this study, we empirically examine the question of whether creditor litigation (external monitoring intervention) is related to low pay-performance sensitivity (PPS), an important indicator of weak corporate governance.

We choose PPS as our indicator of weak corporate governance for two reasons. First, compensation schemes are a dominant method by which shareholders aim to restrict management behavior. The effect of compensation schemes on management behavior has long been studied in the accounting and finance literature (Johnson et al., 1993; Jensen and Murphy, 1990; John and John, 1993; Sung, 2005; Brockman et al., 2010). Compensation schemes are the outcome of the Type I agency problem between managers and shareholders. Information asymmetry between management and shareholders creates incentives and opportunities for management to exploit the interests of the company for their own benefits, instead of maximizing shareholder value. Therefore, compensation contracts are designed to align management interests with the objective of maximizing shareholder value (Jensen and Murphy, 1990). Second, by its nature, compensation is designed by the board of directors and shareholders, without the involvement of creditors. For this reason, compensation schemes provide an appropriate setting to test the conflicts between shareholders and creditors, as creditors have no control over these schemes.

Common practices in compensation contracts include performance-based bonuses, salary revisions, stock options, and performance-based dismissal decisions. Among them, PPS is the most important. PPS indicates how a one-dollar change in compensation is related to a one-dollar change in shareholder wealth (Jensen and Murphy, 1990). Previous studies find that PPS is a strong indicator of corporate governance (Harford and Li, 2007; Kumar and Sivaramakrishnan, 2008; Dhole et al., 2015). As return on assets (ROA) is the measure of firm profitability, low PPS suggests that management compensation is not directly related to the maximization of shareholder wealth. Corporate governance is weak in such circumstances because management incentives cannot be reinforced by compensation schemes.

Corporate governance studies generally conclude that internal governance interacts with external governance to protect the rights of stakeholders, in a substitutive or complementary manner. McKinsey's (2000) surveys find that in countries with weak legal systems, firms with good corporate governance receive premi-

ums. Similarly, Chen et al. (2009) find that in emerging markets, country-level shareholder protection plays a substitutive role with firm-level corporate governance. Using auditors' governance functions, Choi and Wong (2007) show that corporate governance is more important when the legal environment of a country is weak. However, other studies find a substitutive rather than complementary relationship between the two governance mechanisms. El Ghoul et al. (2018) find that auditors' governance role is essential for firm debt maturity, but only in countries with a strong legal environment. In other words, previous studies provide mixed evidence of how corporate governance interacts with the legal environment to reduce the two types of moral hazard problem. In particular, limited attention is paid to how internal corporate governance, dominated by shareholders, interacts with external creditor lawsuits when the agency problem between shareholders and creditors is severe. To fill this research gap, we empirically examine this question in this study.

Using creditor litigation data from China, we find that firms that have experienced creditor lawsuits have low PPS before lawsuits, indicating that these firms have weak corporate governance. This result is consistent with the argument that creditors participate in corporate governance by introducing external monitoring, i.e., lawsuits, when the internal monitoring mechanism dominated by shareholders is insufficient. Our results remain robust to different model specifications and after addressing endogeneity problems. We also find that the relationship is stronger for firms with weak internal control, firms in which management holds a higher percentage of shares, and firms that are not audited by a Big 4 audit firm. Moreover, creditor litigation is strongly related to low PPS when external legal enforcement is stronger and in non-state-owned enterprises (non-SOEs). The results of our additional analyses show that firms with weak corporate governance are more likely to experience creditor litigation in the future, which is consistent with the main results.

We use the Chinese context to examine our research question because creditor lawsuits are an important form of external monitoring in China. China is the world's largest emerging market and its capital market has grown rapidly recently. However, the agency problem is severe in Chinese firms, because strong political intervention and a relatively poor information environment increase moral hazard problems. The Chinese legal system is also different from that of other countries. The Chinese government has stressed the importance of the rule of law since the beginning of the 21st century. However, the legal environment in China still needs to be improved. Allen et al. (2005) demonstrate that the degree of protection of the rights of creditors in China lies between that in countries of English origin (high protection) and that in countries of French origin (weak protection). They also find a low level of law enforcement in China, accompanied by a very high level of corruption. La Porta et al. (2004) show that the protection of property rights in China is weak by global standards, as is political and economic freedom. Therefore, the effect of the legal environment on China's rapidly growing economic and financial systems is worth exploring.

Another important aspect of China's capital market is the dominant role of creditors in providing financing. In China, the main source of firm financing is debt from financial institutions. However, creditors need to monitor firms to protect their rights. Creditors, as corporate outsiders, can participate little in corporate governance, which is dominated by shareholders; instead, they have to monitor firms through litigation (external monitoring). However, creditors face obstacles. The effectiveness of creditor monitoring relies heavily on the legal and bankruptcy systems. If the legal system does not efficiently identify contract violations and does not provide the means to declare bankruptcy and reorganize firms, creditors lose a crucial mechanism of corporate governance (Levine, 2004). Traditionally, bankruptcy regulations have sought to protect the rights of creditors by preventing shareholders from shifting risk to them. The bankruptcy system in China is not properly regulated and enforced, and even provides shelter for firms to avoid paying creditors. In such circumstances, creditors must rely on external monitoring, the legal system, to protect their rights. The negative signal sent by a lawsuit increases the financing constraints of firms. As debt financing is the main source of financing for Chinese firms, creditor lawsuits are devastating. Once a lawsuit is initiated, the negative signal prevents other creditors from providing financing to a firm. Therefore, creditor litigation is of great importance to Chinese firms. We use this setting to examine how external monitoring introduced by creditors interacts with internal monitoring, which is dominated by shareholders.

Our study makes several contributions to the literature. First, it explains how external monitoring based on the legal environment interacts with internal corporate governance. Previous studies mainly focus on the question of whether country-level legal protection and firm-level corporate governance play substitutive or complementary roles. Unlike previous studies that examine whether firm-level monitoring works and take country-level monitoring as a given, our study examines whether the legal system intervenes when corporate governance is weak. This question is rarely studied in the literature.

Second, our study contributes to the literature on the agency problem, especially the conflicts between shareholders and creditors. Conflicts between shareholders and creditors are common in companies, particularly for Chinese firms that use debt capital to form an optimal capital structure. Unlike previous studies focusing on a single type of agency problem, we study the conflicts between creditors and shareholders when the agency problem between the shareholders and managers is severe. On the one hand, because creditors receive a fixed interest rate on their investment, they are risk-averse and prefer not to invest in risky projects once their earnings are sufficient to pay back their investment. However, shareholders invest in the company to maximize their market prices, so they prefer risky projects for higher returns. On the other hand, management acts as the agent of shareholders, not creditors. In addition, directors decide on executive compensation schemes on behalf of the shareholders. Therefore, managers are unlikely to be motivated to make decisions in the interest of creditors, and it is difficult for creditors to participate in the managerial decision-making process. When corporate governance, dominated by shareholders, is inefficient, as suggested by low PPS, creditors need to introduce external monitoring to protect their rights. Therefore, we contribute to the literature by examining the conflicts between shareholders and creditors and their different roles in corporate governance.

Third, we contribute to the literature on corporate governance in China. The legal environment and corporate governance in China, as the world's largest emerging market, are very different from those in other countries. Previous studies in law and economics generally exclude China from their cross-country analysis due to data limitations and China's unique economic and political environment. We extend the literature by providing evidence that the monitoring mechanism of the legal system in China plays a substitutive role and is introduced by creditors when corporate governance is weak.

The rest of the study is organized as follows. Section 2 provides a literature review and presents the hypotheses. Section 3 describes the sample and the variables. Section 4 presents the main results and the robustness tests. Section 5 discusses the results of the cross-sectional analysis. Finally, Section 6 concludes the study.

2. Literature review and hypothesis development

2.1. Literature review

2.1.1. Legal environment and creditor litigation

The law and economics literature generally concludes that investor protection varies between countries (La Porta et al., 1998; La Porta et al., 2000; Djankov et al., 2008). It shows that legal protection is influenced by institutional factors, such as economic development, culture, and firm political connections. Most of these studies consider a country's legal system as exogenous and examine its effect on corporate governance (Pagano and Volpin, 2005). However, laws and regulations are products of the political process, which is essentially the product of economic development. Economic development also influences corporate governance. Therefore, how the legal system and corporate governance interact is an empirical question.

La Porta et al. (2000) find that external protection through the legal system is important because corporate insiders, especially controlling shareholders and managers, are likely to expropriate the interests of creditors and small shareholders. Therefore, small shareholders protect themselves from expropriation by participating in corporate governance. Klapper and Love (2004) find that the country-level legal environment varies across countries and that corporate governance is weaker in countries with a weaker legal environment. Hoskisson et al. (2009) show that stricter monitoring leads to higher CEO compensation. In addition, Acharya et al. (2011) find that external governance and internal governance complement each other. Using survey data, Misangyi and Acharya (2014) show that firms perform better when CEO incentive alignment and monitoring mechanisms work together, complementing rather than substituting for each other. They also show an increase in profit when both internal and external monitoring mechanisms are present. However, monitoring mechanisms obviously combine in complex ways, such that there may be simultaneous substitution and complementarity between and within the various monitoring and control mechanisms. Overall, previous studies

find that institutional factors, such as the legal environment, affect corporate governance and can serve as external monitoring when internal monitoring is insufficient.

Creditor lawsuits are the most common type of stakeholder litigation in China. Creditor litigation increases firm risks and affects external financing. The prosecution of a company by creditors means that the stable creditor-borrower relationship has broken down, which will greatly increase the risks of the business and harm the interests of shareholders. After creditor litigation, creditors that are more sensitive to credit will inevitably adjust a firm's credit conditions and may even be reluctant to make loans. This will worsen the firm's financial conditions, resulting in a shortage of funds, which will affect internal cash flow and asset liquidity. In addition, according to legal procedures, once a lawsuit is filed, the corresponding claims will be kept and some of the assets of the company will be frozen, unable to be traded or used normally. Therefore, creditor litigation is a powerful tool for creditors to intervene in corporate governance.

2.1.2. The two types of agency problem

There are two types of agency problem. Type I exists when there is a divergence of interests between managers (the agent) and firm owners (the principal) (Jensen and Meckling, 1976; Jensen and Murphy, 1990). Type II exists between shareholders and creditors (John and John, 1993; John et al., 2010). In the first type, shareholders expect management to invest in value maximization projects. However, managers may prioritize their own benefits and costs when making decisions. In this case, they may expropriate the value of the owners for their own interests. To align the interests of management with those of shareholders, compensation schemes are often used (Jensen and Meckling, 1976; Jensen and Murphy, 1990).

However, it is understood that shareholder value maximization does not necessarily lead to welfare maximization. Decisions that increase shareholder value may result in costs for creditors. In the second type of agency problem, the roles of creditors and shareholders are different in terms of business activities. Creditors expect their investment and interest to be repaid. Shareholders expect their market prices to be maximized. Therefore, creditors get fixed interest from their investment by using debt contracts. They also have priority of repayment in the event of bankruptcy. However, increased leverage imposes additional risks on firms, and in particular on shareholders. In contrast, shareholders do not get a fixed interest rate on their capital investment, so they prefer high dividends and increased share prices. Accordingly, shareholders may choose riskier, high-return projects. In contrast, creditors do not benefit from these premiums, so investing in risky projects is less attractive. In short, creditors are more interested in value preservation, while shareholders are more interested in value maximization. This conflict can induce agency costs, such as excessive dividend payments, claim dilution, asset substitution, underinvestment or overinvestment, and excessive covenants in loan contracts (Jensen and Meckling, 1976; Smith and Warner, 1979; Macey, 1991; Becht et al., 2003; Chu, 2017). Therefore, the prevailing view in the accounting and finance literature is that shareholder maximization is the second best solution to the problem of corporate governance.

Most studies focus on the first type of agency problem. Among the various compensation mechanisms, performance-based bonuses and dismissal decisions are the most used. By linking CEO incentives with financial performance, CEOs are motivated to boost accounting numbers, which is at the heart of shareholder value maximization. PPS is therefore an indicator of corporate governance. Previous studies generally conclude that higher PPS indicates better corporate governance (Morck et al., 1988; McConnell and Servaes, 1990; Jensen and Murphy, 1990). For instance, Conyon and He (2011) find that in China, compensation is positively related to firm performance and that PPS is higher in Chinese firms with independent directors and in non-SOEs. Minnick et al. (2011) show that in bank holding companies with high PPS, managers make better acquisition decisions. Similarly, Chahine and Goergen (2014) reveal that higher PPS improves IPO performance. They also find that the positive effect of social ties on IPOs is more pronounced when PPS is higher and that the negative effect of family ties on IPOs is mediated by high PPS. Dai et al. (2014) show that in the event of litigation, PPS decreases, and that after litigation, PPS increases They attribute these results to a negative relationship between CEO incentives and firm risks.

However, the literature rarely discusses how compensation schemes reflect and influence shareholder-creditor conflicts; in other words, how the two types of agency problems interact. When the first type of agency problem, i.e., conflicts between shareholders and managers, becomes severe, due to the weak role of creditors in internal governance, creditors must take legal action. Therefore, lawsuits are an important

tool for creditors to intervene in governance. Jiang et al. (2010) find that a reduction in conflict between shareholders and debtholders reduces loan yield spreads. Liao (2015) argues that monitoring by blockholders exacerbates the conflict between debt and equity, in turn affecting the choice and structure of debt financing. Chu (2017) find that firms with more conflicts between equity holders and debtholders pay more dividends, suggesting a transfer of wealth from creditors to shareholders. The two exceptions in this line of research are John and John (1993) and John et al. (2010). Using a theoretical model, John and John (1993) find that performance-based compensation can mitigate shareholder-creditor conflicts. As the leverage ratio increases, PPS should decrease to stop shareholders from shifting risk to creditors, thereby protecting their rights. Similarly, John et al. (2010) examine the model empirically and find a negative relationship between PPS and the leverage ratio. They also show that PPS increases if there is strict monitoring by outsiders of firms' risky investments.

2.2. Hypothesis development

As corporate governance is dominated by shareholders, and compensation schemes are put in place by shareholders to mitigate the first type of agency problem, it is difficult for creditors to participate directly in corporate governance. Previous studies use debt covenant violations or bankruptcy to examine how creditors intervene in corporate governance in extreme cases. As it is difficult to obtain debt covenant and violation data, we extend the research question by using a unique setting, creditor lawsuits, in which external governance by creditors is clearly anticipated by shareholders and managers.

Creditor lawsuits can reflect corporate governance in two ways. First, in companies with weak corporate governance, both types of agency problem are severe. Indeed, self-interested managers consider their gains and losses first when making investment decisions. The risk of overinvestment or loss of return due to underinvestment reduces shareholder value. Second, creditors are more risk-averse than shareholders, because they only receive a fixed payment regardless of share prices. However, they may not be able to get their payment back in the event of firm bankruptcy. Therefore, creditors are concerned about weak corporate governance.

Based on the literature, shareholders establish performance-based compensation plans to link CEOs' actions with their own interests. With high PPS, CEOs are incentivized to act to increase firm performance, thereby increasing shareholder value. In firms with low PPS, CEOs are less motivated by accounting numbers and more likely to expropriate the interests of shareholders for their own benefit, suggesting a weak corporate governance environment. This weak corporate governance due to the agent-principal problem further aggravates conflict between shareholders and creditors, because creditors are more risk-averse than shareholders. When creditors fear that they will not receive the principal and interest due to uncertainty resulting from the risky actions taken by managers, they need to intervene with external governance, via the legal system.

Based on the above discussion, our first hypothesis is as follows:

H1: Firms with creditor lawsuits have low PPS before lawsuits.

Next, we examine whether the interaction between the external legal environment and internal corporate governance varies for firms with different levels of agency problem. The relationship between creditor lawsuits and PPS is not necessarily the same across firms, and the likelihood of external lawsuits may be directly related to the severity of the agency problem. To the extent that a severe agency problem between shareholders and creditors dampens corporate governance, we expect to find low PPS before lawsuits in firms with a more serious agency problem; that is, more serious conflicts between shareholders and creditors. In line with this argument, Huang (2009) finds that creditors tend to impose a mandatory repayment covenant called an "excess cash flow sweep" in loan contracts to force borrowers to repay their debt ahead of schedule when creditorshareholder conflicts are severe. Chu (2017) shows that an increase in the difficulty of class action lawsuits reduces loan spreads and that this effect is weaker in countries where creditors have more rights. In addition, Qi et al. (2011) find that firms with stronger shareholder control face an increase in shareholder-creditor conflicts. Therefore, we expect the relationship between creditor lawsuits and PPS to be stronger in firms with more severe creditor-shareholder conflicts. Our second hypothesis is as follows:

H2: The negative relationship between creditor lawsuits and PPS is more pronounced in firms with more severe creditor-shareholder conflicts.

Numerous studies provide evidence that the behavior of creditors and shareholders is shaped by the legal environment, especially legal institutions for the protection of creditors. For example, Brockman and Unlu (2009) find that country-level creditor rights influence dividend policies by balancing power between debt and equity claimants. They also show that compared with the agency cost of equity, the agency cost of debt plays a more decisive role in dividend policies. Consistent with this argument, Qi et al. (2011) find that creditors require fewer covenants in contracts in countries with stronger creditor protection, indicating that creditor protection replaces covenants in reducing the agency cost of debt. Overall, these studies generally suggest that creditors consider the influence of the external legal environment when intervening in external monitoring. We predict that a mature and strong legal environment will enable creditors to use lawsuits to protect themselves. Therefore, creditors are more likely to participate in corporate governance in the form of lawsuits when creditor protection is strong. Our third hypothesis is as follows:

H3: The negative relationship between creditor lawsuits and PPS is more pronounced in stronger legal environments.

3. Research design

3.1. Model specification

3.1.1. Following the literature (Cadman et al., 2010), we test the relationship between creditor lawsuits and corporate governance using the following specification:

Compensation_{it} =
$$\alpha_0 + \alpha_1 LIT_{it+1} + \alpha_2 ROA_{it} + \alpha_3 ROA_{it} * LIT_{it+1} + Controls + Industry F.E.$$

+ Year F.E. + ε (1)

$$Compensation_{it} = \alpha_0 + \alpha_1 LIT_{it+1} + \alpha_2 ROE_{it} + \alpha_3 ROE_{it} * LIT_{it+1} + Controls + Industry \ F.E.$$

$$+ Year \ F.E. + \varepsilon,$$
(2)

where i and t are the firm and year indicators, respectively. Based on previous studies (Cadman et al., 2010), we measure $Compensation_{it}$ as the natural logarithm of the sum of the cash salaries of the top three highest paid executives. In China, compensation is commonly paid in cash salary. LIT_{it+1} is a dummy variable equal to one if firms have creditor lawsuits in year t+1, and zero otherwise. Firm performance is proxied by ROA_{it} or ROE_{it} (Firth et al., 2006). As we examine how creditors intervene in weak corporate governance, we use the lead time t+1 to investigate how weak corporate governance in the current period results in creditor intervention in the future. Therefore, the interaction term of ROA_{it}/ROE_{it} and LIT_{it+1} represents the relationship between corporate governance and the likelihood of creditor lawsuits. We choose financial performance metrics rather than market performance metrics, such as stock prices and returns, for the following reason. In China, there are many more individual investors than institutional investors, which is very different from the situation in developed markets. Retail investors are less rational and less able to collect and process fundamental information than institutional investors (Barber and Odean, 2008; Barber et al., 2009). Thus, stock prices cannot fully reflect the performance of Chinese listed firms. As a result, it is better to use financial performance based compensation measures in the context of China (Fang, 2009; Wang and Zhang, 2012).

We also include a set of control variables used to determine compensation in the literature (Cornett et al., 2008; Fang, 2009). Specifically, we control for $Size_{it}$, Lev_{it} , $Growth_{it}$, $Fshr_{it}$, $Mshr_{it}$, $Bsize_{it}$, $IndBoard_{it}$, $\Delta Sales_{it}$, $RetVol_{it}$, SOE_{it} , and $Dual_{it}$. $Size_{it}$ is measured as the natural logarithm of a firm's total assets. Lev_{it} is total liabilities scaled by total assets. $Growth_{it}$ is measured as the annual change ratio of the firm's total assets. $Fshr_{it}$ is the percentage of outstanding shares held by the firm's largest shareholder. $Mshr_{it}$ is the percentage of shares

held by the firm's executives. $Bsize_{it}$ is the size of the board, measured by the number of directors on the board. $IndBoard_{it}$ is board independence, which is the percentage of the total number of directors on the board represented by independent directors. $\Delta Sales_{it}$ is the annual change in total revenue. $RetVol_{it}$ is the standard deviation of weekly returns in year t. SOE_i is an indicator variable equal to one if the firm is a state-owned enterprise (SOE), and zero otherwise. Finally, $Dual_{it}$ is an indicator variable equal to one if the chairman and the CEO are the same person in year t. All continuous variables are winsorized at the top and bottom 1%. Year and industry fixed effects are included to control for unobservable factors that are invariant across years and industries. All standard errors are robust to heteroskedasticity and clustered at the firm level.

3.2. Sample construction

Our initial sample is based on all A-share firms listed on the Shenzhen Stock Exchange and the Shanghai Stock Exchange. We manually collect lawsuit data from corporate annual reports and construct a proprietary database of creditor litigation. The sampling period extends from 2003 to 2013. We obtain firm financial data from the China Stock Market & Accounting Research database. All financial firms are excluded because the financial sector is heavily regulated. We also remove all firms with missing financial and stock market data. Our final sample includes 12,321 firm-year observations from 1943 firms. Our main variable of interest is LIT_{it+1} , a dummy variable equal to one for all firm-year observations with creditor lawsuits, and zero otherwise. Therefore, our treatment firms include all firm-year observations with creditor litigation, while our control firms include all firm-year observations with no creditor litigation.

3.3. Descriptive statistics

The distribution of the sample is presented in Table 1. Column (1) presents the annual distribution of firms without litigation (non-litigation firms). Column (2) shows the annual distribution of firms with litigation (litigation firms). We observe an upward trend in the number of lawsuits, from 48 lawsuits in 2003 to 123 lawsuits in 2013. We have 1058 firm-year observations in the treatment group and 11,263 firm-year observations in the control group.

Table 2 presents the summary statistics of the final sample. The mean ratio of LIT_{it+1} is 0.086, suggesting that about 8.6% of the firms in our sample faced litigation during the sampling period. The mean of *Compensation*_{it} is 13.732, with a standard deviation of 0.852. Therefore, there is a wide variation in the amount of compensation between firms. The mean value of ROA_{it} is 0.034 and the mean value of ROE_{it} is 0.066. The mean ratio of $Size_{it}$ is 21.7, with a median of 21.568. The other statistics are generally comparable to those reported in previous studies.

Table 1 Sample Distribution.

| Year | $LIT_{t+I} = 0$ | $LIT_{t+1} = 1$ | Total |
|-------|-----------------|-----------------|--------|
| 2003 | 226 | 48 | 274 |
| 2004 | 401 | 68 | 469 |
| 2005 | 700 | 91 | 791 |
| 2006 | 871 | 94 | 965 |
| 2007 | 895 | 117 | 1012 |
| 2008 | 980 | 119 | 1099 |
| 2009 | 1101 | 101 | 1202 |
| 2010 | 1195 | 92 | 1287 |
| 2011 | 1512 | 89 | 1601 |
| 2012 | 1688 | 116 | 1804 |
| 2013 | 1694 | 123 | 1817 |
| Total | 11,263 | 1058 | 12,321 |

Note: In this table shows the distribution of the sample by year.

Table 2 Descriptive Statistics.

| Variable | N | M | SD | Min | P25 | Median | P75 | Max |
|----------------------------|-------|--------|-------|--------|--------|--------|--------|--------|
| $\overline{LIT_{it+1}}$ | 12321 | 0.086 | 0.280 | 0.000 | 0.000 | 0.000 | 0.000 | 1.000 |
| Compensation _{it} | 12321 | 13.732 | 0.852 | 9.932 | 13.214 | 13.780 | 14.305 | 16.047 |
| ROA_{it} | 12321 | 0.034 | 0.077 | -0.946 | 0.010 | 0.036 | 0.069 | 0.255 |
| ROE_{it} | 12321 | 0.066 | 0.204 | -3.023 | 0.024 | 0.077 | 0.140 | 1.464 |
| $Size_{it}$ | 12321 | 21.700 | 1.253 | 18.657 | 20.849 | 21.568 | 22.399 | 25.925 |
| Lev_{it} | 12321 | 0.504 | 0.271 | 0.043 | 0.334 | 0.498 | 0.643 | 3.678 |
| $Growth_{it}$ | 12321 | 0.177 | 0.382 | -0.716 | 0.008 | 0.101 | 0.234 | 3.786 |
| $Fshr_{it}$ | 12321 | 0.366 | 0.156 | 0.081 | 0.240 | 0.346 | 0.486 | 0.758 |
| $Mshr_{it}$ | 12321 | 0.028 | 0.089 | 0.000 | 0.000 | 0.000 | 0.000 | 0.577 |
| $Bsize_{it}$ | 12321 | 9.151 | 1.855 | 5.000 | 9.000 | 9.000 | 9.000 | 15.000 |
| $IndBoard_{it}$ | 12321 | 0.363 | 0.050 | 0.143 | 0.333 | 0.333 | 0.375 | 0.571 |
| $\Delta Sales_{it}$ | 12321 | 0.224 | 0.620 | -0.885 | -0.015 | 0.130 | 0.303 | 7.087 |
| $RetVol_{it}$ | 12321 | 0.134 | 0.057 | 0.035 | 0.096 | 0.121 | 0.159 | 0.671 |
| SOE_i | 12321 | 0.554 | 0.497 | 0.000 | 0.000 | 1.000 | 1.000 | 1.000 |
| Dual _{it} | 12321 | 0.179 | 0.383 | 0.000 | 0.000 | 0.000 | 0.000 | 1.000 |

Note: In this table presents the descriptive statistics for the main variables used in the regressions.

4. Pay-Performance sensitivity and creditor lawsuits

4.1. Main analysis

Table 3 presents the results of our main analysis based on Eqs. (1) and (2). In column (1) and column (2), in which firm performance is proxied by ROA, we find significant negative coefficients for the interaction term of ROA and LIT. Specifically, in column (1), no control variables are included and the coefficient of ROA_{it}^* - LIT_{it+1} is significantly negative (-1.691, t = -5.71). In column (2), all of the control variables are included and the coefficient of ROA_{it}^* - LIT_{it+1} remains significantly negative (-1.869, t = -7.11). In columns (3) and (4), in which firm performance is proxied by ROE, the coefficients of ROE_{it}^* - LIT_{it+1} remain significantly negative. Specifically, in column (3), no control variable is included and the coefficient of ROE_{it}^* - LIT_{it+1} is -0.922, with a t-value of -9.79. Similarly, in column (4), the coefficient of ROE_{it}^* - LIT_{it+1} is -0.538, with a t-value of -6.28. The results are consistent with our expectations. In firms with high PPS, managers are incentivized by high salaries and motivated to improve corporate governance. Therefore, corporate governance is better in firms with high PPS, suggesting fewer conflicts between creditors and shareholders. As a result, creditors are less likely to participate in corporate governance in the form of lawsuits.

The signs of the coefficients of the control variables are generally consistent with those of previous studies (Firth et al., 2006; Chen et al., 2015). We find a positive relationship between compensation and firm size, the percentage of management shareholdings, board size, board independence, whether the firm is an SOE, and whether the CEO of the firm is also the chairman of the board. We find a negative association between compensation and leverage, firm growth, the percentage of shares held by the largest shareholder, annual sales growth, and stock return volatility.

In summary, the results in Table 3 support H1 postulating that firms with creditor lawsuits have low PPS before lawsuits. Indeed, there is a negative relationship between creditor lawsuits and PPS, suggesting that creditors are more likely to participate in corporate governance via an external litigation approach when corporate governance dominated by shareholders is weak.

4.2. Robustness Checks

We check the robustness of our main results using a series of sensitivity analyses. We first examine whether our main results are robust to an alternative litigation measure. Specifically, we originally defined LIT_{it+1} as whether firms have experienced litigation. When collecting the data, we observed that many firms have more than one lawsuit. Therefore, we changed our definition of creditor litigation to $LITnum_{it+1}$, measured by the number of lawsuits in which firms have been involved, i.e., the frequency of lawsuits. The results are presented

Table 3 Main Results.

| Variable | (1) | (2) | (3) | (4) |
|-------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|-----------|-----------|-----------|-------------------|
| $\overline{LIT_{it+1}}$ | -0.107*** | 0.001 | -0.209*** | -0.002 |
| | (-2.99) | (0.02) | (-5.37) | (-0.06) |
| ROA_{it} | 3.481*** | 2.864*** | , , | · · · · · · · · · |
| | (19.79) | (15.99) | | |
| $ROA_{it} * LIT_{it+1}$ | -1.691*** | -1.869*** | | |
| | (-5.71) | (-7.11) | | |
| ROE_{it} | ` , | , , | 1.118*** | 0.771*** |
| | | | (13.11) | (10.62) |
| $ROE_{it} * LIT_{it+1}$ | | | -0.922*** | -0.538*** |
| | | | (-9.79) | (-6.28) |
| Size _{it} | | 0.289*** | , | 0.304*** |
| u | | (23.94) | | (24.83) |
| Lev_{it} | | -0.066 | | -0.330*** |
| | | (-1.26) | | (-6.16) |
| Growth _{it} | | -0.055*** | | -0.034* |
| n in the second | | (-3.12) | | (-1.94) |
| Fshr _{it} | | -0.314*** | | -0.291*** |
| 1 3 11 | | (-3.66) | | (-3.34) |
| Mshr _{it} | | 0.185 | | 0.247** |
| 1/15/10/11 | | (1.56) | | (2.04) |
| Bsize _{it} | | 0.028*** | | 0.029*** |
| Botzen | | (3.73) | | (3.83) |
| IndBoard _{it} | | 0.133 | | 0.091 |
| Inaboura _{ll} | | (0.60) | | (0.40) |
| $\Delta Sales_{it}$ | | -0.040*** | | -0.024** |
| ZDuics _{it} | | (-3.76) | | (-2.19) |
| $RetVol_{it}$ | | -0.185 | | -0.248 |
| Tet v or _{it} | | (-1.22) | | (-1.58) |
| SOE_{it} | | 0.004 | | -0.007 |
| SOL_{it} | | (0.13) | | (-0.23) |
| Dual _{it} | | 0.097*** | | 0.093*** |
| Duu_{it} | | (3.01) | | (2.83) |
| Constant | 12.304*** | 6.184*** | 12.332*** | 6.006*** |
| Constant | (110.94) | (22.88) | (110.43) | (21.96) |
| Ind & Year | YES | YES | YES | YES |
| Observations | | | | |
| R^2 | 12,321 | 12,321 | 12,321 | 12,321 |
| Λ | 0.336 | 0.479 | 0.308 | 0.465 |

Note: In this table presents the results of the main regression. The figures reported in brackets are t-statistics, based on standard errors clustered at the firm level. ***, **, and * indicate significance at the 1%, 5%, and 10% levels, respectively.

in Panel A, Table 4. Our main results remain unchanged. In columns (1) and (2), the coefficients of ROA_{it}^* - $LITnum_{it+1}$ are still negative, and equal to -1.030, with a t-value of -9.45, when all controls are included (column 2). The results are similar in columns (3) and (4), with a coefficient of $ROE_{it}^*LITnum_{it+1}$ equal to -0.251 and a t-value of -6.43 (column 4). Therefore, our main results are not sensitive to this alternative measure of creditor litigation.

We also examine whether our main results hold with a different sampling period. A number of laws and enforcement measures relating to creditor protection were introduced in around 2006, such as the Bankruptcy Law and the Property Law. Previous studies show that the adoption of these laws has significantly increased legal protection, especially the protection of creditors (Berkowitz et al., 2015). To eliminate the concern that our results are driven by the adoption of these laws, we use a sampling period after 2006 and re-estimate the equations. The results are reported in Panel B of Table 4. Our results remain unchanged. We still find a negative coefficient for the interaction term of ROA/ROE and LIT. Therefore, the negative relationship between PPS and creditor litigation still holds despite the adoption of legal protection laws.

Finally, we examine whether our results are robust to different regression methods. In the main analysis, all coefficients and standard errors are clustered at the firm level. We change the clustering method to both the

Table 4
Robustness Checks.

| Panel A: Using the number | of lawsuits | | | |
|------------------------------|--------------------|-----------|---------------------|---------------------|
| Variable | (1) | (2) | (3) | (4) |
| $\overline{LITnum_{it+1}}$ | -0.124*** | -0.054* | -0.194*** | -0.029 |
| | (-4.08) | (-1.79) | (-6.41) | (-0.96) |
| ROA_{it} | 3.486*** | 2.799*** | | |
| | (20.57) | (16.03) | | |
| $ROA_{it} * LITnum_{it+1}$ | -1.144*** | -1.030*** | | |
| | (-10.63) | (-9.45) | | |
| ROE_{it} | | | 1.051*** | 0.717*** |
| | | | (13.59) | (10.98) |
| $ROE_{it} * LITnum_{it+1}$ | | | -0.476*** | -0.251*** |
| | | | (-10.26) | (-6.43) |
| Controls | NO | YES | NO | YES |
| Constant | 12.314*** | 6.245*** | 12.337*** | 6.017*** |
| | (111.35) | (23.18) | (110.99) | (22.06) |
| Ind & Year | YES | YES | YES | YES |
| Observations | 12,321 | 12,321 | 12,321 | 12,321 |
| R^2 | 0.339 | 0.480 | 0.308 | 0.465 |
| Panel B: Using the sample | after 2006 | | | |
| Variable | (1) | (2) | (3) | (4) |
| - | | | | |
| LIT_{it+1} | -0.096** | 0.015 | -0.167*** | 0.013 |
| | (-2.55) | (0.46) | (-4.16) | (0.39) |
| ROA_{it} | 3.479*** | 2.855*** | | |
| $ROA_{it} * LIT_{it+1}$ | (17.39) | (13.96) | | |
| | -1.030*** | -1.326*** | | |
| | (-3.10) | (-4.25) | | |
| ROE_{it} | | | 1.417*** | 0.942*** |
| | | | (15.62) | (11.48) |
| $ROE_{it} * LIT_{it+1}$ | | | -0.922*** | -0.477*** |
| | | | (-8.01) | (-4.48) |
| Controls | NO | YES | NO | YES |
| Constant | 13.702*** | 7.492*** | 13.749*** | 7.371*** |
| | (114.87) | (26.26) | (117.18) | (25.88) |
| Ind & Year | YES | YES | YES | YES |
| Observations | 9822 | 9822 | 9822 | 9822 |
| R^2 | 0.229 | 0.405 | 0.213 | 0.394 |
| Panel C: Clustered at the fi | rm and year levels | | | |
| Variable | (1) | (2) | (3) | (4) |
| $\overline{LIT_{it+1}}$ | -0.107*** | 0.001 | -0.209*** | -0.002 |
| LII_{it+1} | (-3.36) | (0.02) | (-5.00) | (-0.06) |
| ROA_{it} | 3.481*** | 2.864*** | (-3.00) | (-0.00) |
| KOA_{il} | (17.22) | (15.72) | | |
| $ROA_{it} * LIT_{it+1}$ | -1.691*** | -1.869*** | | |
| $KOA_{it} \cdot LII_{it+1}$ | | | | |
| ROE_{it} | (-4.30) | (-5.50) | 1.118*** | 0.771*** |
| KOE_{it} | | | | |
| $ROE_{it} * LIT_{it+1}$ | | | (7.21) -0.922*** | (8.12) -0.538*** |
| $ROE_{it} \cdot LII_{it+1}$ | | | | |
| Controls | NO | VEC | (-10.45) | (-8.13) |
| Controls | NO | YES | NO | YES |
| Constant | 13.702*** | 7.492*** | 13.749*** | 7.371*** |
| X 1.0 X7 | (114.87) | (26.26) | (117.18) | (25.88) |
| Ind & Year | YES | YES | YES | YES |
| Observations | 12,321 | 12,321 | 12,321 | 12,321 |
| R^2 | 0.336 | 0.479 | 0.308 | 0.465 |

Note: In this table presents the results of the robustness tests. The figures reported in brackets are t-statistics, based on standard errors clustered at the firm level. ***, **, and * indicate significance at the 1%, 5%, and 10% levels, respectively.

firm level and the year level. The results are reported in Panel C of Table 4. The negative relationship between PPS and creditor litigation remains unchanged. Therefore, our results are robust to different regression methods.

4.3. Endogeneity

Endogeneity may be a concern. There are two potential endogeneity problems in our setting. First, whether creditors initiate litigation is a decision. If there are certain omitted factors that are related to creditors' litigation decisions and these factors affect PPS, endogeneity is a concern. Second, there may be fundamental differences between our treatment sample and control sample. In other words, litigation firms may be fundamentally different in terms of corporate governance or financial reporting from non-litigation firms. We assume that there is no difference between the two groups in our main analysis. If this assumption does not hold, endogeneity is a concern.

To eliminate potential selection bias and omitted variable bias, we use a propensity score matching (PSM) approach (Rosenbaum and Rubin, 1983; Conyon and He, 2016). Specifically, we match litigation firms and non-litigation firms, and re-estimate Eqs. (1) and (2) using the treatment sample and the matched samples.

In the first stage, following the literature (Wang and Jiang, 2016), we include the following variables: $Size_{it}$, Lev_{it} , ROA_{it} , $\Delta sales_{it}$, ST_{it} , PPE_{it} , WC_{it} , Age_{it} , $RetVol_{it}$, and $Zscore_{it}$. ROA_{it} is income before extraordinary items divided by total assets. ST_{it} is an indicator variable equal to one if the share is specially treated in year t. PPE_{it} is net property, plant, and equipment scaled by total assets. WC_{it} is working capital divided by total assets, where working capital is measured as the difference between current assets and current liabilities. Age_{it} is measured by the difference between year t and the firm's listing year. Finally, $Zscore_{it}$ is the bankruptcy index calculated following Altman (1968). The dependent variable is LIT_{it+1} . The other variables are defined in the same way as in the main regression. We use a probit model to estimate the propensity scores and match the treatment firms with the control firms based on the closest propensity score (without replacement) in the same industry and the same year. After identifying matching control firms for each treatment firm, we reestimate Eqs. (1) and (2) using the matched samples.

The results are presented in Table 5. In column (1) and column (2), in which performance is proxied by ROA, the coefficients of $ROA_{it}*LIT_{it+1}$ are still significantly negative, suggesting that firms with low PPS are more likely to be sued in the next period. In column (3) and column (4), in which firm performance is proxied by ROE, the results remain unchanged. Overall, our results remain unchanged when using the PSM method and are not driven by endogeneity.

5. Cross-Sectional variation in the relationship between creditor litigation and Pay-Performance sensitivity

5.1. The role of shareholder-creditor agency conflict

Corporate finance theory suggests that agency conflict is the main source of incentive problems. There are various types of agency costs, including the agency cost of equity (conflicts between equity holders) and the agency cost of debt (conflicts between debtholders and shareholders). In firms with a high agency cost of equity, managers may deviate from the initial objective of firm value maximization. The PPS incentive system may work differently in firms with a severe agency problem. Managers may manipulate firm performance to secure higher salaries, which is difficult to detect when corporate governance is weak. As a result, it is even more difficult for creditors to participate in corporate governance. In contrast, in firms in which the agency cost of debt is high, creditors have more incentives to participate in corporate governance and protect their rights. Therefore, we postulate in H2 that the negative relationship between PPS and creditor lawsuits is stronger in firms with a serious agency problem.

¹ On April 22, 1998, the Shanghai and Shenzhen stock exchanges announced that firms with abnormal financial status should trade with "ST" status. Firms that report losses for two consecutive years are classified as having abnormal financial status.

² The results of the first stage regression are presented in Appendix A1. Appendix A2 presents the matching efficiency results for the main variables used in the PSM approach. The results show that the treatment sample is comparable to the control sample.

Table 5 PSM Results.

| Variable | (1) | (2) | (3) | (4) |
|-----------------------------|-----------|-----------|-----------|-----------|
| LIT_{it+1} | -0.001 | 0.014 | -0.052 | 0.011 |
| | (-0.02) | (0.38) | (-1.19) | (0.31) |
| ROA_{it} | 2.555*** | 1.899*** | | · · · |
| | (8.91) | (7.14) | | |
| $ROA_{it} * LIT_{it+1}$ | -0.921** | -1.037*** | | |
| | (-2.54) | (-3.19) | | |
| ROE_{it} | | | 0.439*** | 0.369*** |
| | | | (4.45) | (5.15) |
| $ROE_{it} * LIT_{it+1}$ | | | -0.278** | -0.183** |
| | | | (-2.52) | (-2.07) |
| Size _{it} | | 0.289*** | ` , | 0.302*** |
| | | (13.94) | | (14.73) |
| Lev _{it} | | -0.045 | | -0.193*** |
| ···· | | (-0.70) | | (-3.12) |
| Growth _{it} | | -0.042 | | -0.011 |
| ** | | (-1.02) | | (-0.26) |
| Fshr _{it} | | -0.377** | | -0.398** |
| | | (-2.45) | | (-2.57) |
| Mshr _{it} | | 0.956*** | | 0.962*** |
| 1.25111 [[| | (2.74) | | (2.76) |
| $Bsize_{it}$ | | 0.031*** | | 0.033*** |
| Butzell | | (2.68) | | (2.88) |
| IndBoard _{it} | | 0.225 | | 0.238 |
| InaBoara _{ll} | | (0.63) | | (0.66) |
| $\Delta Sales_{it}$ | | -0.037* | | -0.029 |
| abares _{it} | | (-1.74) | | (-1.34) |
| $RetVol_{it}$ | | -0.481* | | -0.442 |
| Kei v Oi _{it} | | (-1.77) | | (-1.57) |
| SOE_{it} | | 0.012 | | 0.013 |
| SOL_{il} | | (0.24) | | (0.26) |
| Dual _{it} | | 0.102* | | 0.096 |
| Duui _{it} | | (1.73) | | (1.63) |
| Constant | 12.309*** | 6.135*** | 12.319*** | 5.905*** |
| Constant | | | | |
| I. J 0. V | (75.77) | (13.29) | (72.50) | (12.84) |
| Ind & Year | YES | YES | YES | YES |
| Observations p ² | 2114 | 2114 | 2114 | 2114 |
| R^2 | 0.337 | 0.468 | 0.297 | 0.462 |

Note: In this table presents the results of propensity score matching. The figures reported in brackets are *t*-statistics, based on standard errors clustered at the firm level. ***, **, and * indicate significance at the 1%, 5%, and 10% levels, respectively.

To test H2, we follow the literature (Kanodia and Lee, 1998; Johnson et al., 2000; Ruland and Zhou, 2005; Rani et al., 2012; Chen et al., 2015) and build proxies for the agency problem between shareholders and creditors. Specifically, we use three proxies. The first measure is internal control weakness (*ICW*). The weaker the internal control system, the greater the agency problem between shareholders and creditors. The second proxy is the percentage of shares held by management (*Mshr*). Firms with a higher percentage of management shareholdings are perceived as having a more severe agency problem between shareholders and creditors. The third proxy is whether the firm is audited by a Big 4 audit firm (*Big 4*). Big 4 auditors strictly control financial information, which reduces the agency problem. Therefore, creditors are less concerned about corporate governance in firms with Big 4 auditors. For each agency problem proxy, we divide the full sample into firms with a high agency problem and firms with a low agency problem based on the sample median.

The results are reported in Table 6. Panel A presents the results using ICW as the agency problem proxy. In the subsample with low ICW, the coefficient of $ROA_{it}*LIT_{it+1}$ is 1.145, with a t-value of 1.12. In the subsample with high ICW, the coefficient of $ROA_{it}*LIT_{it+1}$ is -1.736, with a t-value of -6.75. The results are similar when using ROE as the firm performance proxy in column (3) and column (4). We continue to find more pronounced results in the high agency problem subsample using other proxies (Panel B and Panel C). Overall, the

Table 6
The Role of Shareholder-Creditor Agency Conflict.

| Panel A: High versus low ICW | | | | |
|------------------------------------------------------------------------------------------------------------------------|----------------------|---------------------|----------------------|--------------------|
| Variable | (1) High | (2) Low | (3) High | (4) Low |
| LIT _{it+1} | -0.051 (-1.43) | 0.001 (0.01) | -0.032 (-0.88) | 0.129* (1.72) |
| ROA_{it} | 2.218*** (11.97) | 3.523*** (10.79) | (-0.88) | (1.72) |
| $ROA_{it} * LIT_{it+1}$ | -1.736*** (-6.75) | 1.145 (1.12) | | |
| ROE_{it} | (-0.73) | (1.12) | 0.474*** (7.51) | 1.571*** (9.91) |
| $ROE_{it} * LIT_{it+1}$ | | | -0.313*** (-4.13) | -0.638 (-1.25) |
| Controls | YES | YES | YES | YES |
| Constant | 6.702*** | 6.218*** | 6.489*** | 6.452*** |
| | (19.12) | (17.80) | (18.30) | (18.38) |
| Ind & Year | YES | YES | YES | YES |
| Observations | 6100 | 093 | 6100 | 6093 |
| R ² | 0.458 | 0.432 | 0.445 | 0.427 |
| Diff-Test for the coefficients of $ROA_{it} * LIT_{it+1}$ Diff-Test for the coefficients of $ROE_{it} * LIT_{it+1}$ | | $\chi^2 = 8.31***$ | | $\chi^2 = 4.58**$ |
| Panel B: High versus low Mshr | (1) | (2) | (2) | |
| Variable | (1) | (2) | (3) | |
| | High | Low | High | Low |
| LIT_{it+1} | 0.076* | -0.048 | 0.085* | -0.051 |
| PO 4 | (1.69) | (-1.14) $2.423***$ | (1.92) | (-1.21) |
| ROA_{it} | 3.110*** (12.18) | (10.34) | | |
| $ROA_{it} * LIT_{it+1}$ | -1.926*** | -1.716*** | | |
| ROII _{II} EIII _{II} +I | (-4.47) | (-5.26) | | |
| ROE_{it} | () | () | 0.960*** | 0.616*** |
| | | | (7.98) | (7.53) |
| $ROE_{it} * LIT_{it+1}$ | | | -0.626*** | -0.450*** |
| | ***** | ***** | (-4.42) | (-4.46) |
| Controls | YES | YES | YES | YES |
| Constant | 5.420*** | 6.873*** | 5.300*** | 6.716*** |
| Ind & Year | (13.83) YES | (20.99) YES | (12.98) YES | (20.39) YES |
| Observations | 6157 | 6164 | 6157 | 6164 |
| R^2 | 0.504 | 0.464 | 0.490 | 0.455 |
| Diff-Test for the coefficients of $ROA_{it} * LIT_{it+1}$ | | $\chi^2 = 2.78*$ | | |
| Diff-Test for the coefficients of $ROE_{it} * LIT_{it+1}$ | | | | $\chi^2 = 3.02*$ |
| Panel C: Big 4 versus non-Big 4 | | | | |
| Variable | (1) Big 4 | (2) Non-Big 4 | (3) Big 4 | (4) Non-Big 4 |
| LIT_{it+1} | 0.020 | 0.027 | -0.037 | 0.024 |
| | (0.16) | (0.81) | (-0.32) | (0.70) |
| ROA_{it} | 3.297*** | 3.012*** | | |
| | (3.62) | (13.63) | | |
| $ROA_{it} * LIT_{it+1}$ | 1.669 | -1.818*** | | |
| POE | (0.73) | (-5.45) | O C 41** | 0.955*** |
| ROE_{it} | | | 0.641** (2.39) | (11.97) |
| $ROE_{it} * LIT_{it+1}$ | | | 0.563 | -0.556*** |
| | | | (1.05) | (-5.43) |
| | | | | ued on next page) |

Table 6 (continued)

| Variable | (1) | (2) | (3) | (4) |
|-----------------------------------------------------------|----------|------------------|----------|-------------------|
| | Big 4 | Non-Big 4 | Big 4 | Non-Big 4 |
| Controls | YES | YES | YES | YES |
| Constant | 7.998*** | 6.640*** | 7.888*** | 6.771*** |
| | (7.17) | (20.92) | (6.95) | (20.93) |
| Ind & Year | YES | YES | YES | YES |
| Observations | 668 | 9514 | 668 | 9513 |
| R^2 | 0.530 | 0.388 | 0.517 | 0.380 |
| Diff-Test for the coefficients of $ROA_{it} * LIT_{it+1}$ | | $\chi^2 = 3.79*$ | | |
| Diff-Test for the coefficients of $ROE_{it} * LIT_{it+1}$ | | | | $\chi^2 = 4.63**$ |

Note: In this table presents the results for the role of shareholder-creditor agency conflict. The figures reported in brackets are *t*-statistics, based on standard errors clustered at the firm level. ***, ***, and * indicate significance at the 1%, 5%, and 10% levels, respectively.

regression results support H2. Creditors have more incentives to participate in corporate governance via external litigation when the agency problem between shareholders and creditors is more severe.

5.2. The role of the external legal environment

Previous studies generally find that the legal environment plays an important role in shaping the behavior of creditors (Clemenz and Gugler, 2000). For example, Brockman and Unlu (2009) find that country-level creditor rights influence firm-level dividend policies by establishing a balance of power between debtholders and shareholders. The interaction between creditors and managers/shareholders changes with external measures to protect creditors. To the extent that the cost of litigation is lower and litigation is more efficient in a stronger legal environment, we expect the negative relationship between creditor litigation and PPS to be more pronounced in a strong legal environment (H3).

We use two proxies for the legal environment. The first is the legal index. Following Fan and Wang (2012), we use the number of lawyers as a percentage of the population, the efficiency of local courts, and the protection of property rights. The sample is divided into firms with a strong legal environment and firms with a weak legal environment based on the sample median. The second proxy is whether the company is an SOE. It is difficult for external creditors to sue SOEs because of their political affiliation. Therefore, the legal protection of creditors is weaker if they lend to SOEs.

The results are presented in Table 7. In Panel A in which the rigor of the legal environment is proxied by the legal index, we find that the coefficients of $ROA_{it}/ROE_{it}*LIT_{it+1}$ are more significantly negative in the subsample with a strong legal environment. The results are the same when we use SOE status as a proxy (Panel B). The legal environment is stronger in non-SOEs, because political interventions are heavier for SOEs. Overall, the results support H3. In a strong legal environment, the low cost and high efficiency of litigation help creditors participate in corporate governance in the form of lawsuits, so the relationship between creditor lawsuits and PPS is more pronounced in a strong legal environment.

5.3. Alternative measure of corporate governance

In this section, we present our results using an alternative measure of corporate governance: earnings opacity. Previous studies show that corporate governance is strongly related to the information environment of firms (Armstrong et al., 2012). Therefore, we use *EarningsOpacity* as an alternative measure of corporate governance. Greater earnings opacity indicates a poorer information environment and weaker corporate governance.

The results are presented in Table 8. We use three measures of earnings opacity. The first measure is the absolute value of discretionary accruals based on the modified Jones model (Dechow et al., 1995). The second measure is the absolute value of discretionary accruals following Dechow and Dichev (2002). The third measure is earnings opacity, calculated as the three-year moving sum of the absolute value of annual discretionary accruals (Hutton et al., 2009). We find a significantly positive association between earnings opacity and creditor litigation in the next period. The results are consistent with the main results, showing that firms with weaker internal corporate governance are more likely to be subject to creditors' intervention through external lawsuits.

Table 7
The Role of the External Legal Environment.

| Panel A: High versus low Legal Index | (1) | (2) | (2) | 7.4N |
|------------------------------------------------------------------------------------------------------------------------|----------------------|----------------------|----------------------|-----------------------------|
| Variable | (1) High | (2) Low | (3) High | (4) Low |
| LIT_{it+1} | 0.025 (0.62) | 0.009 (0.22) | 0.014 (0.33) | 0.01 ⁴ (0.34) |
| ROA_{it} | 2.619*** (11.15) | 3.017*** (12.52) | () | (111 |
| $ROA_{it} * LIT_{it+1}$ | -2.087*** (-5.72) | -1.745*** (-5.31) | | |
| ROE_{it} | | , | 0.779*** (7.24) | 0.778*** (9.04 |
| $ROE_{it} * LIT_{it+I}$ | | | -0.569*** (-4.60) | -0.533*** (-4.79) |
| Controls | YES | YES | YES | YES |
| Constant | 7.211*** | 5.934*** | 7.180*** | 5.642*** |
| | (20.28) | (17.19) | (20.03) | (16.09) |
| Ind & Year | YES | YES | YES | YES |
| Observations | 5791 | 6530 | 5791 | 6530 |
| R^2 | 0.429 | 0.524 | 0.415 | 0.511 |
| Diff-Test for the coefficients of $ROA_{it} * LIT_{it+1}$ Diff-Test for the coefficients of $ROE_{it} * LIT_{it+1}$ | | $\chi^2 = 3.01^*$ | | $\chi^2 = 2.89^*$ |
| Panel B: Non-SOE versus SOE | | | | |
| Variable | (1) | (2) | (3) | |
| | Non-SOE | SOE | Non-SOE | SOE |
| LIT_{t+1} | -0.018 (-0.36) | 0.003 (0.08) | -0.008 (-0.17) | -0.008 (-0.21) |
| ROA_{it} | 2.763*** | 2.759*** | (, | , |
| " | (10.87) | (11.21) | | |
| $ROA_{it}^* LIT_{t+1}$ | -2.319*** | -0.654* | | |
| | (-7.07) | (-1.89) | | |
| $ROEi_t$ | | | 0.695*** | 0.742*** |
| | | | (6.24) | (9.07) |
| $ROE_{it}^* LIT_{t+1}$ | | | -0.605*** | -0.378*** |
| | | | (-4.56) | (-3.42) |
| Controls | YES | YES | YES | YES |
| Constant | 5.155*** | 6.592*** | 4.921*** | 6.402*** |
| | (11.27) | (19.88) | (10.60) | (19.14) |
| | | YES | YES | YES |
| Ind & Year | YES | IES | | |
| Observations | YES 5499 | 6822 | 5499 | 6822 |
| | | | | 6822 0.516 |

Note: In this table presents the results of the role of the external legal environment. The figures reported in brackets are t-statistics, based on standard errors clustered at the firm level. *** and * indicate significance at the 1% and 10% levels, respectively.

6. Conclusion

In this study, we investigate whether creditors can participate in corporate governance when agency conflict between shareholders and creditors is severe. We examine our research question using creditor litigation data from China, comparing litigation firms and non-litigation firms. We find that firms that have experienced creditor lawsuits have low PPS before lawsuits, suggesting that these firms have weak corporate governance. This result is consistent with the argument that creditors participate in corporate governance by introducing external monitoring when internal monitoring dominated by shareholders is insufficient. We also find that this relationship is stronger in firms with more severe agency conflict between shareholders and creditors. Moreover, creditor litigation is strongly related to low PPS when the external legal environment is stronger. Our results

Table 8 Alternative Measure of Corporate Governance.

| Dep. Var: Lit _{it+1} | (1) | (2) | (3) |
|--------------------------------|---------------|-----------|-----------|
| | Abs_DA_adjust | Abs_DA_DD | Opacity |
| Earnings Opacity _{it} | 0.150*** | 0.201*** | 0.072*** |
| | (3.96) | (4.29) | (3.06) |
| $Size_{it}$ | -0.026*** | -0.028*** | -0.033*** |
| | (-6.79) | (-7.10) | (-7.75) |
| ROA_{it} | -0.408*** | -0.391*** | -0.386*** |
| | (-6.33) | (-5.93) | (-5.33) |
| Lev_{it} | 0.217*** | 0.221*** | 0.219*** |
| | (10.08) | (9.94) | (9.19) |
| $Growth_{it}$ | -0.025*** | -0.022** | -0.006 |
| | (-2.66) | (-2.10) | (-0.60) |
| $Fshr_{it}$ | -0.055** | -0.055** | -0.044* |
| | (-2.54) | (-2.41) | (-1.67) |
| $Mshr_{it}$ | -0.077*** | -0.090** | -0.115 |
| | (-2.68) | (-2.57) | (-1.55) |
| $Bsize_{it}$ | 0.001 | 0.002 | 0.003 |
| | (0.61) | (0.88) | (1.42) |
| $\Delta Sales_{it}$ | 0.009 | 0.010 | 0.009 |
| | (1.52) | (1.62) | (1.34) |
| $RetVol_{it}$ | 0.293*** | 0.292*** | 0.296*** |
| | (4.32) | (4.11) | (3.72) |
| SOE_{it} | -0.018** | -0.019** | -0.022** |
| | (-2.04) | (-2.05) | (-2.23) |
| $Dual_{it}$ | 0.003 | 0.003 | 0.001 |
| | (0.38) | (0.27) | (0.10) |
| Constant | 0.642*** | 0.687*** | 0.788*** |
| | (7.66) | (7.84) | (8.19) |
| Ind & Year | YES | YES | YES |
| Observations | 12,024 | 11,164 | 9396 |
| R^2 | 0.125 | 0.126 | 0.124 |

Note: In this table presents the results of the alternative measure of corporate governance. The figures reported in brackets are *t*-statistics, based on standard errors clustered at the firm level. ***, **, and * indicate significance at the 1%, 5%, and 10% levels, respectively.

remain robust to different model specifications, after addressing endogeneity problems, and to the use of an alternative measure of corporate governance.

Our study contributes to the literature in several ways. First, it expands the literature on the interaction between external monitoring based on the legal environment and internal corporate governance. In particular, our study shows that the legal protection of creditors is effective when corporate governance is weak. Second, our study contributes to the literature on the agency problem, in particular on conflicts between shareholders and creditors in China, where this type of agency problem is common. Finally, we contribute to the literature on accounting and law in China's capital market. The legal environment and corporate governance in China, as the largest emerging market, are very different from those in other countries. We extend the literature by providing evidence that monitoring by the legal system in China plays a key role and is effectively introduced by creditors when corporate governance is weak. Our results also have implications to policymakers in that legal environment not only directly affects firm behaviors, but also influences creditors' role in corporate governance.

Declaration of Competing Interest

The authors declare that they have no known competing financial interests or personal relationships that could have appeared to influence the work reported in this paper.

Acknowledgements

We gratefully acknowledge helpful comments from Hongping Duan (discussant), Erik Johannesson (discussant), conference participants at the 2018 China Empirical Accounting Conference, the 2018 2nd China Finance and Accounting Research Symposium at Xiamen University, and the 2019 Accounting and Business Research (ABR) & China Journal of Accounting Research (CJAR) Joint Conference. We acknowledge financial support from the National Natural Science Foundation of China (#71572210; #71802205) and the Fundamental Research Funds for the Central Universities (#SWU1909771).

Appendix A

See Tables A1 and A2

Table A1
The First-Stage Regression Results of PSM.

| Variable | (1) |
|---------------------|--------------------|
| Size _{it} | -0.148*** |
| | (-8.17) |
| Lev_{it} | 0.682*** |
| | (8.21) |
| ROA_{it} | -1.891*** |
| 46.1 | (-7.67) |
| $\Delta Sales_{it}$ | 0.005 |
| ST_{it} | (0.21) 0.517*** |
| S1 it | (8.08) |
| PPE_{it} | -0.369*** |
| и | (-3.05) |
| WC_{it} | 0.000 |
| | (0.01) |
| Age_{it} | 0.291*** |
| | (8.36) |
| $RetVol_{it}$ | 1.645*** |
| 7 | (4.42) |
| $Zscore_{it}$ | -0.013*** (-2.63) |
| Constant | 1.503*** |
| Constant | (3.59) |
| Ind & Year | YES |
| Observations | 12,241 |
| R^2 | 0.178 |

Note: In this table presents the first-stage regression results of PSM. The figures reported in brackets are *t*-statistics, based on standard errors clustered at the firm level. *** indicates significance at the 1% level.

Table A2 Matching Efficiency for the Variables Used in PSM.

| Variable | LIT = 0 | LIT = 1 | Diff_Test (p value) |
|----------------|---------|---------|---------------------|
| Size | 21.325 | 21.321 | 0.999 |
| Lev | 0.626 | 0.637 | 0.103 |
| ROA | 0.006 | -0.017 | 0.000*** |
| $\Delta Sales$ | 0.261 | 0.210 | 0.088* |
| PPE | 0.261 | 0.262 | 0.579 |
| WC | -0.012 | -1.020 | 0.185 |
| Age | 2.371 | 2.376 | 0.591 |
| RetVol | 0.152 | 0.153 | 0.518 |

Note: In this table presents the mean values and the difference in mean values for the matching variables between the treatment sample and the control sample. *** and * indicate significance at the 1% and 10% levels, respectively.

References

- Abdi, M., Aulakh, P.S., 2012. Do country-level institutional frameworks and interfirm governance arrangements substitute or complement in international business relationships? J. Int. Busin. Stud. 43 (5), 477–497.
- Acharya, V.V., Myers, S.C., Rajan, R.G., 2011. The internal governance of firms. J. Finan. 66 (3), 689-720.
- Allen, F., Qian, J., Qian, M., 2005. Law, finance, and economic growth in China. J. Financ. Econ. 77 (1), 57-116.
- Altman, E.I., 1968. Financial ratios, discriminant analysis and the prediction of corporate bankruptcy. J. Finan. 23 (4), 589-609.
- Appel, I., 2019. Governance by litigation. https://doi.org/10.2139/ssrn.2532278.
- Armour, J., Menezes, A., Uttamchandani, M., Zwieten, K., 2015. How do creditor rights matter for debt finance? A review of empirical evidence. Edward Elgar Publishing, pp. 3–25.
- Armstrong, C.S., Balakrishnan, K., Cohen, D., 2012. Corporate governance and the information environment: Evidence from state antitakeover laws. J. Account. Econom. 53 (1–2), 185–204.
- Barber, B.M., Lee, Y.T., Liu, Y.J., Odean, T., 2009. Just how much do individual investors lose by trading? Rev. Finan. Stud. 22 (2), 609–632.
- Barber, B.M., Odean, T., 2008. All that glitters: The effect of attention and news on the buying behavior of individual and institutional investors. Rev. Finan. Stud. 21 (2), 785–818.
- . Corporate governance and control. In Handbook of the economics of finance Vol. 1, 1-109.
- Berkowitz, D., Lin, C., Ma, Y., 2015. Do property rights matter? Evidence from a property law enactment. J. Financ. Econ. 116 (3), 583–593.
- Brockman, P., Martin, X., Unlu, E., 2010. Executive compensation and the maturity structure of corporate debt. J. Finan. 65 (3), 1123–1161.
- Brockman, P., Unlu, E., 2009. Dividend policy, creditor rights, and the agency costs of debt. J. Financ. Econ. 92 (2), 276-299.
- Cadman, B., Carter, M.E., Hillegeist, S., 2010. The incentives of compensation consultants and CEO pay. J. Account. Econom. 49 (3), 263–280.
- Chahine, S., Goergen, M., 2014. Top management ties with board members: How they affect pay-performance sensitivity and IPO performance. J. Corporate Finan. 27, 99–115.
- Chen, K.C., Chen, Z., Wei, K.J., 2009. Legal protection of investors, corporate governance, and the cost of equity capital. J. Corporate Finan. 15 (3), 273–289.
- Chen, S., Lin, B., Lu, R., Zhang, T., 2015. Controlling shareholders' incentives and executive pay-for-performance sensitivity: Evidence from the split share structure reform in China. J. Int. Finan. Markets, Inst. Money 34, 147–160.
- Choi, J.H., Wong, T.J., 2007. Auditors' governance functions and legal environments: An international investigation. Contemp. Account. Res. 24 (1), 13–46.
- Chu, Y., 2017. Shareholder litigation, shareholder-creditor conflict, and the cost of bank loans. J. Corporate Finan. 45, 318–332.
- Clemenz, G., Gugler, K., 2000. Macroeconomic development and civil litigation. Eur. J. Law Econ. 9 (3), 215-230.
- Conyon, M.J., He, L., 2011. Executive compensation and corporate governance in China. J. Corporate Finan. 17 (4), 1158–1175.
- Conyon, M.J., He, L., 2016. Executive compensation and corporate fraud in China. J. Bus. Ethics 134 (4), 669-691.
- Cornett, M.M., Marcaus, A.J., Tehranian, H., 2008. Corporate governance and pay-for-performance: The impact of earnings management. J. Financ. Econ. 87 (2), 357–373.
- Dai, Z., Jin, L., Zhang, W., 2014. Executive pay-performance sensitivity and litigation. Contemp. Account. Res. 31 (1), 152-177.
- Dechow, P.M., Dichev, I.D., 2002. The quality of accruals and earnings: The role of accrual estimation errors. Account. Rev. 77 (1), 35–59.
- Dechow, P.M., Sloan, R.G., Sweeney, A.P., 1995. Detecting earnings management. Account. Rev. 70 (2), 193-225.
- Dhole, S., Khumawala, S.B., Mishra, S., Ranasinghe, T., 2015. Executive compensation and regulation-imposed governance: Evidence from the California Nonprofit Integrity Act of 2004. Account. Rev. 90 (2), 443–466.
- Djankov, S., La Porta, R., Lopez-de-Silanes, F., Shleifer, A., 2008. The law and economics of self-dealing. J. Financ. Econ. 88 (3), 430–465.
- El Ghoul, S., Guedhami, O., Kim, Y., 2017. Country-level institutions, firm value, and the role of corporate social responsibility initiatives. J. Int. Busin. Stud. 48 (3), 360–385.
- El Ghoul, S., Guedhami, O., Lennox, C.S., & Pittman, J., 2018. External versus internal monitoring: The importance of multiple large shareholders and families to auditor choice in Western European firms. http://doi.org/10.2139/ssrn.1373808.
- Ernstberger, J., Grüning, M., 2013. How do firm- and country-level governance mechanisms affect firms' disclosure? J. Account. Public Policy 32 (3), 50–67.
- Fan, G., & Wang, X. (2012). The report on the relative process of marketization of each region in China. The Economic Science Press. (in Chinese).
- Fang, J., 2009. Is top management compensation of Chinese public companies sticky? Econ. Res. J. 3, 110-124 (in Chinese).
- Firth, M., Fung, P.M., Rui, O.M., 2006. Corporate performance and CEO compensation in China. J. Corporate Finan. 12 (4), 693-714.
- Gale, D., Hellwig, M., 1985. Incentive-compatible debt contracts: The one-period problem. Rev. Econ. Stud. 52 (4), 647-663.
- Gungoraydinoglu, A., Öztekin, Ö., 2011. Firm-and country-level determinants of corporate leverage: Some new international evidence. J. Corporate Finan. 17 (5), 1457–1474.
- Harford, J., Li, K., 2007. Decoupling CEO wealth and firm performance: The case of acquiring CEOs. J. Finan. 62 (2), 917-949.
- Hart, O., Moore, J., 1998. Default and renegotiation: A dynamic model of debt. Quart. J. Econ. 113 (1), 1-41.

Hoskisson, R.E., Castleton, M.W., Withers, M.C., 2009. Complementarity in monitoring and bonding: More intense monitoring leads to higher executive compensation. Acad. Manage. Perspect. 23 (2), 57–74.

Huang, R., 2009. Creditor control of free cash flow. http://doi.org/10.2139/ssrn.1502957.

Hutton, A.P., Marcus, A.J., Tehranian, H., 2009. Opaque financial reports, R2, and crash risk. J. Financ. Econ. 94 (1), 67-86.

Jensen, M.C., Meckling, W.H., 1976. Theory of the firm: Managerial behavior, agency costs and ownership structure. J. Financ. Econ. 3 (4), 305–360.

Jensen, M.C., Murphy, K.J., 1990. Performance pay and top-management incentives. J. Polit. Econ. 98 (2), 225-264.

Jiang, G., Lee, C.M., Yue, H., 2010. Tunneling through intercorporate loans: The China experience. J. Financ. Econ. 98 (1), 1-20.

John, K., Mehran, H., Qian, Y., 2010. Outside monitoring and CEO compensation in the banking industry. J. Corporate Finan. 16 (4), 383–399.

John, T.A., John, K., 1993. Top-management compensation and capital structure. J. Finan. 48 (3), 949-974.

Johnson, R.A., Hoskisson, R.E., Hitt, M.A., 1993. Board of director involvement in restructuring: The effects of board versus managerial controls and characteristics. Strateg. Manag. J. 14 (S1), 33–50.

Johnson, S., Boone, P., Breach, A., Friedman, E., 2000. Corporate governance in the Asian financial crisis. J. Financ. Econ. 58 (1–2), 141–186.

Kanodia, C., Lee, D., 1998. Investment and disclosure: The disciplinary role of periodic performance reports. J. Account. Res. 36 (1), 33–55.

Kim, K.A., Kitsabunnarat-Chatjuthamard, P., Nofsinger, J.R., 2007. Large shareholders, board independence, and minority shareholder rights: Evidence from Europe. J. Corporate Finan. 13 (5), 859–880.

Klapper, L.F., Love, I., 2004. Corporate governance, investor protection, and performance in emerging markets. J. Corporate Finan. 5 (10), 703–728.

Kroszner, R.S., Strahan, P.E., 2001. Bankers on boards: Monitoring, conflicts of interest, and lender liability. J. Financ. Econ. 62 (3), 415–452

Kumar, P., Sivaramakrishnan, K., 2008. Who monitors the monitor? The effect of board independence on executive compensation and firm value. Rev. Finan. Stud. 21 (3), 1371–1401.

La Porta, R., Lopez-de-Silanes, F., Pop-Eleches, C., Shleifer, A., 2004. Judicial checks and balances. J. Polit. Econ. 112 (2), 445-470.

La Porta, R., Lopez-de-Silanes, F., Shleifer, A., Vishny, R., 2000. Investor protection and corporate governance. J. Financ. Econ. 58 (1), 3–27.

La Porta, R., Lopez-de-Silanes, F., Shleifer, A., Vishny, R.W., 1998. Law and finance. Journal of Political Economy 106 (6), 1113–1155. Levine. R., 2004. The corporate governance of banks: A concise discussion of concepts and evidence. World Bank. The.

Liao, S., 2015. Outside blockholders' monitoring of management and debt financing. Contemp. Account. Res. 32 (4), 1373-1404.

Macey, J.R., 1991. An economic analysis of the various rationales for making shareholders the exclusive beneficiaries of corporate fiduciary duties. Stetson Law Review 21, 23.

Martynova, M., Renneboog, L., 2008. Spillover of corporate governance standards in cross-border mergers and acquisitions. J. Corporate Finan. 14 (3), 200–223.

McConnell, J.J., Servaes, H., 1990. Additional evidence on equity ownership and corporate value. J. Financ. Econ. 27 (2), 595-612.

Minnick, K., Unal, H., Yang, L., 2011. Pay for performance? CEO compensation and acquirer returns in BHCs. Rev. Finan. Stud. 24 (2), 439-472.

Misangyi, V.F., Acharya, A.G., 2014. Substitutes or complements? A configurational examination of corporate governance mechanisms. Acad. Manag. J. 57 (6), 1681–1705.

Morck, R., Shleifer, A., Vishny, R.W., 1988. Management ownership and market valuation: An empirical analysis. J. Financ. Econ. 20, 293–315.

Nini, G., Smith, D.C., Sufi, A., 2012. Creditor control rights, corporate governance, and firm value. Rev. Finan. Stud. 25 (6), 1713–1761. Pagano, M., Volpin, P.F., 2005. The political economy of corporate governance. Am. Econ. Rev. 95 (4), 1005–1030.

Qi, Y., Roth, L., Wald, J.K., 2011. How legal environments affect the use of bond covenants. J. Int. Busin. Stud. 42 (2), 235-262.

Rani, H., Udi, H., Johnstone, K.M., 2012. Internal control material weakness and CFO compensation. Contemp. Account. Res. 29 (3), 768–803.

Rosenbaum, P.R., Rubin, D.B., 1983. The central role of the propensity score in observational studies for causal effects. Biometrika 70 (1), 41–55.

Ruland, W., Zhou, P., 2005. Debt, diversification, and valuation. Rev. Quant. Financ. Acc. 25 (3), 277-291.

Shleifer, A., Vishny, R.W., 1997. A survey of corporate governance. J. Finan. 52 (2), 737-783.

Smith Jr, C.W., Warner, J.B., 1979. On financial contracting: An analysis of bond covenants. J. Finan. Econ. 7 (2), 117-161.

Sung, J., 2005. Optimal contracts under adverse selection and moral hazard: A continuous-time approach. Rev. Finan. Stud. 18 (3), 1021–1073.

Townsend, R.M., 1979. Optimal contracts and competitive markets with costly state verification. J. Econ. Theory 21 (2), 265-293.

Wang, H., Zhang, R., 2012. Private equity and executive compensation contracts: From the perspective of corporate governance. Manage. World 9, 156–167 (in Chinese).

Wang, Y., Jiang, G., 2016. The effect of litigation on tunneling of debt. Manage. Rev. 28 (1), 191–204 (in Chinese).

Williamson, O.E., 1988. Corporate finance and corporate governance. J. Finan. 43 (3), 567–591.